Author's response to reviews

Title: Intravascular papillary endothelial hyperplasia of the orbit in a Nigerian child: Case report and review of literature

Authors:

Oluyemi Fasina (yemifash2000@yahoo.com)
Adewunmi O Adeoye (wunmiadeoye@gmail.com)
Effiong E Akang (eeakang@yahoo.com)

Version: 2 Date: 9 February 2012

Author's response to reviews:

Title: Intravascular papillary endothelial hyperplasia of the orbit in a Nigerian child: Case report and review of literature
Authors:
FASINA Oluyemi*
Department of Ophthalmology,
University College Hospital
Ibadan,
Oyo State,
Nigeria.
Email: yemifash2000@yahoo.com
Phone: +234 805 403 4787
*Corresponding author

Adewunmi Adeoye FWACP
Department of Pathology,
University College Hospital,
Ibadan,
Nigeria.
Email: wunmiadeoye@gmail.com

Effiong Akang FWACP
Department of Pathology,
University College Hospital,
Ibadan,
Nigeria.
Covering letter
The Editor,
Journal of Medical Case Report
Dear Sir,
Re: Intravascular papillary endothelial hyperplasia of the orbit in a Nigerian child: Case report and review of literature
We hereby present a case of intravascular papillary endothelial hyperplasia of the orbit in a young Nigerian, who underwent orbitotomy and tumor excision. Intravascular papillary endothelial hyperplasia is an uncommon benign tumor, and the orbit is a rare site of affectionation.
Late presentation of patients is still a major problem in our environment, resulting in irreversibly poor vision of the affected eye in our patient, and thus, concerted efforts must be made to enlighten the populace on the dangers of this.
Worthy of note is the fact that this, to the best of our knowledge, is the first case of orbital intravascular papillary endothelial hyperplasia in a child in our environment, contributing to the uniqueness of our case. This article had been previously submitted to BMC Clinical Pathology journal.
Thank you for considering our article in your highly esteemed journal.

Dr Fasina O.